ANSWERING REVIEWERS

November 07, 2016



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 30194-review.doc).

Title: Foot and Ankle History and Clinical Examination: A Guide to Everyday Practice

Author: Sulaiman Alazzawi, Mohamed Sukeik, Daniel King, Krishna Vemulapalli,

Name of Journal: World Journal of Orthopedics

ESPS Manuscript NO: 30194

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 References and typesetting were corrected

Thank you again for publishing our manuscript in the World Journal of Orthopedics

Sincerely yours,

Sulaiman Alazzawi

Trauma and Orthopaedic Department, Royal London Hospital, London E1 1BB, United Kingdom